**Author’s response to reviews**

**Title:** Better Palliative Care for people with a Dementia: Summary of Interdisciplinary Workshop Highlighting Current Gaps and Recommendations for Future Research

**Authors:**

Siobhan Fox (s.fox@ucc.ie)

Carol FitzGerald (carol.fitzgerald@ucc.ie)

Karen Harrison Dening (Karen.Harrison-Dening@dementiauk.org)

Kate Irving (kate.irving@dcu.ie)

W. George Kernohan (wg.kernohan@ulster.ac.uk)

Adrian Treloar (adrian.treloar@nhs.net)

David Oliver (drdjoliver@gmail.com)

Suzanne Guerin (suzanne.guerin@ucd.ie)

Suzanne Timmons (s.timmons@ucc.ie)

**Version: 2 Date: 23 Jun 2017**

**Author’s response to reviews:**

Please see revised version attached, without tracked changes as requested. Regards, Siobhan Fox